Computer Science CSC 238H Wednesday July 15 1998 St. George Campus University of Toronto

Homework Assignment #3 Due: Wednesday August 5 1998 at 6:10pm

Cover Page

Please fill out this page and attach it to the front of your assignment. Without this completed cover page, your assignment will not be marked.

First Name:

Last Name:

Student Number:

Tutorial (circle one):

Joe Geneviève Alex Steve Eugenia SF 1101 GB 244 GB 119 GB 120 SF 3109 (remedial) (Erin, New, PHE) (Trin, Inn, Scar) (Vic, SMC, Eng) (Wood, UC)

Below, please list the names of every student with whom you have discussed the homework as well as the titles of every textbook you consulted (other than the course notes). Then, please sign the cover page to indicate that you have read and understood the policy on Collaboration in Homework stated in the Course Information handout.

Signature:

(Without your signature, your assignment will not be marked.)